Author's response to reviews

Title: Increased health care utilization by survivors of childhood lymphoblastic leukemia is confined to those treated with cranial or total body irradiation: a case cohort study

Authors:

Anna S Holmqvist (anna.sallfors-holmqvist@med.lu.se)
Christian Moëll (christian.moell@skane.se)
Lars Hjorth (lars.hjorth@skane.se)
Anna Lindgren (anna@maths.lth.se)
Stanislaw Garwicz (stanislaw.garwicz@skane.se)
Thomas Wiebe (thomas.wiebe@skane.se)
Ingrid Øra (ingrid.ora@med.lu.se)

Version: 4 Date: 4 June 2014

Author's response to reviews: see over
To BMC Cancer

Dear Editor Dafne Solera and Reviewers,

Thank you very much for the opportunity to copyedit our manuscript entitled “Increased health care utilization of survivors of childhood lymphoblastic leukemia confined to those treated with cranial or total body irradiation: a case cohort study”. Before submission to BMC Cancer, the manuscript was revised by a language editor. After your request for copyediting, the manuscript has been reviewed by a native-English speaking medical researcher.

On behalf of all the authors,

Sincerely,

Anna Sällfors Holmqvist, MD
Pediatric Oncology and Hematology
Clinical Sciences, Lund University
SE-221 85 Lund, Sweden
Phone: +46 46 178294,
Fax: +46 46 172299
E-mail: anna.sallfors-holmqvist@med.lu.se

Ingrid Øra, MD, PhD
Assoc. Prof. / Senior Consultant
Pediatric Oncology and Hematology
Clinical Sciences, Lund University
22185 Lund, Sweden
Phone: +46-722214431
E-mail: ingrid.ora@med.lu.se